Author's response to reviews

Title: Assessment of anti-inflammatory tumor treatment efficacy by longitudinal monitoring employing sonographic micro morphology in a preclinical mouse model.

Authors:
Sanjay Tiwari (stiwari@email.uni-kiel.de)
Jan H Egberts (Jan-Hendrik.Egberts@uksh-kiel.de)
Olena Korniienko (lena_kornienko@email.uni-kiel.de)
Linda Köhler (Linda.Weder@uksh-kiel.de)
Anna Trauzold (atrauzold@email.uni-kiel.de)
Claus C Glüer (glueer@rad.uni-kiel.de)
Holger Kalthoff (hkalthoff@email.uni-kiel.de)

Version: 6 Date: 21 June 2011

Author's response to reviews:

Dear Sir/Madam

Two minor editing changes have been made. (1). One author on the author list did not have a number next to her name to indicate Institute affiliation. This has been corrected.

(2) The email address of the authors was in brackets and underlined. The brackets and the line have been removed.

Yours sincerely
Sanjay Tiwari, PhD